Author's response to reviews

Title: Feasibility and impact of an intensified antibiotic stewardship programme targeting cephalosporin and fluoroquinolone use in a tertiary care university medical center

Authors:

Johannes P Borde (johannes.borde@web.de)
Klaus Kaier (kaier@imbi.uni-freiburg.de)
Michaela Steib-Bauert (michaela.steib-bauert@uniklinik-freiburg.de)
Werner Vach (wv@imbi.uni-freiburg.de)
Annette Geibel-Zehender (anette.geibel-zehender@universitaets-herzzentrum.de)
Hansjörg Busch (hansjoerg.busch@uniklinik-freiburg.de)
Hartmut Bertz (hartmut.bertz@uniklinik-freiburg.de)
Martin Hug (martin.hug@uniklinik-freiburg.de)
Katja de With (katja.dewith@uniklinikum-dresden.de)
Winfried V Kern (winfried.kern@uniklinik-freiburg.de)

Version: 2
Date: 11 January 2014

Author's response to reviews: see over
Dear Editorial Board,

attached I may send you our manuscript with the title “Feasibility and impact of an intensified antibiotic stewardship programme targeting cephalosporin and fluoroquinolone use in a tertiary care university medical center”. Our trial indicates that an intensified ABS programme targeting cephalosporin und fluoroquinolone use in the setting of a large academic hospital is feasible and sustainable. Using interrupted time-series analysis we demonstrate a significant intervention-related decrease in the consumption of cephalosporins and fluoroquinolones (p<0.001) outperforming the decreasing baseline trend. The intervention may serve as a model for other services and hospitals with a similar structure and baseline situation.

The revised submission contains now the required ethics statement and a “Acknowledgements” section.

Kind regards,

Johannes Borde, MD